Author's response to reviews

Title: Synovial chondromatosis originating from the synovium of the anterior cruciate ligament: a case report

Authors:

Tokifumi Majima (tkmajima@med.hokudai.ac.jp)
Tamotsu Kamishima (ktamotamo2@med.hokudai.ac.jp)
Kouichi Susuda (mika.s@s-s-orth-hp.or.jp)

Version: 5 Date: 20 April 2009

Author's response to reviews:

We revised as follows:
Page 2, 2 lines from last: Flexibility was changed to flexion
Page 4, 2 nd para, line 8: "synovium ACL" was changed to " synovium of ACL"